Research Paper



Effect of Opium Versus Methadone on Polysomnographic Characteristics of Patients With Obesity Hypoventilation Syndrome

Babak Amra^{1, 2}, Hossein Narimani², Forogh Soltaninejad^{1, 2*}, Mehrzad Salmasi², Marjan Mansourian³, Thomas Penzel⁴, Christoph Schöbel⁵, Ingo Fietze⁴

- 1. Pulmonary and Sleep Ward, Bamdad Respiratory and Sleep Research Center, Isfahan University of Medical Sciences, Isfahan, Iran.
- 2. Department of Internal Medicine, School of Medicine, Isfahan University of Medical Sciences, Isfahan, Iran.
- 3. Department of Epidemiology and Biostatistics, School of Health, Isfahan University of Medical Sciences, Isfahan, Iran.
- 4. Department of Cardiology and Pulmonology, Sleep Medicine Center, Charité-Berlin University of Medicine, Berlin, Germany.
- 5. Department of Pneumology, West German Lung Center, University Medicine Essen, Essen, Germany.



Citation Amra, B., Narimani, H., Soltaninejad, F., Salmasi, M., Mansourian, M., & Penzel, T., et al. (2023). Effect of Opium Versus Methadone on Polysomnographic Characteristics of Patients With Obesity Hypoventilation Syndrome. *Basic and Clinical Neuroscience*, *15*(1), 101-108. http://dx.doi.org/10.32598/bcn.2022.3901.1

doi http://dx.doi.org/10.32598/bcn.2022.3901.1



Article info: Received: 24 Dec 2021 First Revision: 27 Jan 2022 Accepted: 07 Feb 2022 Available Online: 01 Jan 2024

Keywords:

Polysomnography, Obesity hypoventilation syndrome, Opium dependence Opiate substitution treatment, Surveys, Questionnaires

ABSTRACT

Introduction: There are studies about polysomnographic (PSG) characteristics of patients with either obesity hypoventilation syndrome (OHS) or addiction. We aimed to investigate the PSG characteristics of obstructive sleep apnea (OSA) patients with opium addiction, those on methadone maintenance treatment (MMT), and non-addicts for the treatment of addiction.

Methods: In this cross-sectional study, we enrolled 75 patients with OHS in the Bamdad Respiratory and Sleep Research Center affiliated with the Isfahan University of Medical Sciences between January 2020 and February 2021. The patients were categorized into three groups: Opium addicts (OA), MMT, and non-addicts (NA). All patients completed screening questionnaires for OSA. This included the Epworth sleepiness scale (ESS), stop-bang questionnaire, and Berlin questionnaire and the data analyzed by SPSS software, version 24.

Results: A total of 75 OHS patients (54 men [72%] and 21 women [28%]) were studied in three groups, including OA (n=30), MMT (n=15), and NA (n=30). The apnea hypopnea index was not significantly different between the three groups. The longest apnea duration was higher in the OA than in other groups (P=0.001). Central apnea index (P=0.01), longest hypopnea duration (P=0.04), PaCO₂ (P=0.04), and time with SpO₂<90% (T₉₀) (P=0.009) were higher in the MMT than in other groups. Furthermore, the minimum SpO₂ was lower in the MMT than in other groups (P=0.03).

Conclusion: Some of the sleep disturbances were worse in the MMT than in the OA group. This suggests the need for further studies to compare the effects of opium and methadone on sleep in OHS patients.

* Corresponding Author:

Forogh Soltaninejad, Assistant Professor.

Address: Department of Internal Medicine, School of Medicine, Isfahan University of Medical Sciences, Isfahan, Iran. Tel: +98 (31) 32214047 E-mail: soltaninejad.fg@gmail.com, f.soltaninejad@med.mui.ac.ir

.....

Highlights

• The severity of obstructive sleep apnea (OSA) in obesity hypoventilation syndrome (OHS) patients on methadone maintenance treatment (MMT) is the same as opium addicts.

- Sleep hypoxia is more severe and prolonged in OHS patients on MMT than opium addicts.
- Sleep hypercapnia is more severe in OHS patients on MMT compared to opium addicts.

Plain Language Summary

In our study conducted in Bamdad Respiratory and Sleep Research Center in Isfahan, we investigated the polysomnographic characteristics of three group of patients with confirmed sleep disordered breathing. These groups included of patients with BMI \geq 30, daytime hypercapnia (PCO₂ \geq 45) and sleep disordered breathing and opium addiction, treatment with methadone, or none of them. Polysomnography is the standard method for diagnosis of sleep breathing disorders. In this method, several health records including respiration, cardiac, brain and muscle activity are monitored. Methadone has been the approved treatment for opium addiction. The severity of sleep disordered breathing as presented with apnea hypopnea index was not different between groups. However, we found some sleep disturbances including central apnea, duration of sleep with oxygen saturation less than 90%, and the minimum oxygen saturation were worst in under methadone treatment patients. These findings highlight the urgent need for more research on different aspects of methadone treatment.

1. Introduction

besity hypoventilation syndrome (OHS) is defined by the triad of obesity (body mass index [BMI] \geq 30 kg/m²), daytime hypoventilation (awake $CO_2 \ge 45 \text{ mm Hg}$) and sleep-disordered breathing without an alternative neuromuscular, mechanical, or metabolic cause of hypoventilation (Iftikhar & Roland, 2018). The syndrome is often associated with obstructive sleep apnea (OSA), which causes periods of reduced breathing during sleep, resulting in many arousals and awakenings during the night, which results in daytime sleepiness (Kakazu et al., 2020). The treatment of choice for OHS is positive airway pressure therapy in the form of continuous positive airway pressure (CPAP) or non-invasive ventilation (NIV). CPAP consists of one continuous pressure that prevents OSA, but NIV consists of bi-level pressure, including inspiratory and expiratory pressures for additional ventilatory support (Masa et al., 2019).

Methadone, a long-acting μ -opioid agonist, has been recognized as one of the best treatments for opioid addiction. Methadone maintenance treatment (MMT) is a comprehensive treatment program that involves the long-term administration of methadone as a substitute for the opiate, to which the subject has become addicted. (Le et al., 2019). Nowadays, MMT programs are available in both developing and developed countries under the supervision of physicians (Russolillo et al., 2018).

Opium addiction is common in the Iranian population and has been one of the important health issues in recent years. Based on formal reports, 1.8% of the 15–64-yearold population is addicted to opioids in Iran, and some studies have suggested that this value might be as high as 2.8% of the population over the age of 15 years (Aflatoonian et al., 2014; Ekhtiari et al., 2020). On the other hand, methadone usage prevalence is thought to be <0.6% in the Iranian population (Shekarchizadeh et al., 2012).

Methadone and other opioids have various effects on sleep quality and structure in acute and chronic use. Based on previous studies, compared to healthy individuals, these people had more daily drowsiness, and in terms of polysomnographic (PSG) characteristics, these patients had increased waking time and less REM sleep, and in general, sleep disorders, especially central sleep apnea, was more prevalent in these patients (Baldassarri et al., 2020). It has been demonstrated that 75% of patients under treatment with methadone had an apneahypopnea index (AHI) above five and there was a correlation between AHI and the daily dose of methadone (McPherson et al., 2019; Finlay et al., 2020) There is a complex bidirectional relationship between sleep and substance abuse (Ara et al., 2016). So far, various studies have evaluated the prevalence of sleep disorders in patients addicted to opioids or on MMT (Martel et al., 2018, Mubashir et al., 2020). Furthermore, PSG characteristics of OHS patients have been investigated in recent studies. To the best of our knowledge, very few studies have been conducted on this issue and no studies have evaluated the PSG characteristics of these patients in the Iranian population. Considering the prevalence of OHS and also addiction in our region, it is important to study the PSG characteristics of these patients.

2. Materials and Methods

Study design and participants

This is a cross-sectional observational study that was conducted between January 2020 and February 2021 at the Bamdad Respiratory and Sleep Research Center affiliated with the Isfahan University of Medical Sciences. The current study included patients referred to our sleep laboratory with suspected sleep breathing disorders for a PSG investigation.

The inclusion criteria were BMI of more than 30 kg/m², daily hypercapnia (PaCO₂ \geq 45), confirmed sleep breathing disorder on PSG, and written informed consent to participate in this study. The exclusion criteria were the presence of any known under-treatment neurologic or psychiatric diseases and the consumption of sedative or hypnotic drugs. Inclusion criteria for the OA group were the diagnosis of inhalational opium addiction and for the MMT group, the included patients were on the MMT program for more than three months. Patients were categorized into three groups, including opium addicts (OA), MMT, and non-addicts (NA). The confirmation of addiction was made by an expert psychiatrist.

Data collection

Demographic data of all patients, including age, gender, and past medical histories were collected. All patients completed screening questionnaires for OSA. This included the epworth sleepiness scale (ESS), stop-bang questionnaire, and Berlin questionnaire under the guidance of trained personnel. Standard attended overnight PSG was performed on all patients, and the following variables were monitored: Electroencephalography, electro-oculography, chin and leg electromyography, electrocardiography, oxygen saturation, respiratory effort, oronasal airflow and snoring sounds, and body position (Soltaninejad et al., 2017). Scoring of all PSGs was performed by a certified somnologist according to the American Academy of Sleep Medicine (AASM) scoring manual 2.6 (Jorgensen et al., 2020). The same equipment was used for titration of positive pressure treatment CPAP, bi-level positive airway pressure, and bi-level positive airway pressure with backup rate.

PSG data included AHI, mean SpO₂, minimum SpO₂, central apnea index (CAI), longest apnea, longest hypopnea, and sleep duration with SpO₂<90%.

Statistical analysis

The statistics analysis was performed using SPSS software, version 24. Non-numerical variables were reported as numbers and percentages and numerical variables as Mean±SD or median and mid-quarter amplitude. Comparison of groups was performed based on quantitative variables using analysis of variance (ANOVA) and quantitative variables were compared using the Kruskal-Wallis-test. Linear and nominal multilevel regression analysis was used to evaluate the effect of treatment in three groups by eliminating the effect of possible confounders. All analyses were performed considering a 5% error level.

3. Results

Demographic characteristics

A total number of 75 OHS patients (54 men [72%] and 21 women [28%]) were enrolled in three groups (OA (30), MMT (15), and NA (30)) based on inclusion and exclusion criteria. The mean age of patients was 55.73 ± 14.47 years and the mean BMI in patients was 37.11 ± 6.73 kg/m².

The demographic characteristics of the three groups are shown in Table 1. Among the demographic characteristics, the groups were significantly different only regarding gender (P=0.001). The scores of all OSA screening questionnaires including ESS, stop-bang questionnaire, and Berlin questionnaire were higher in OA patients, although it was non-significant for the Berlin questionnaire (Table 1). Hyperlipidemia was the only comorbidity with significant differences between the groups (P=0.043).

PSG characteristics

The various PSG characteristics of patients in three groups are presented in Table 2. Some variables with significant differences between the groups were CAI, the REM latency, the longest apnea and hypopnea duration,



NEURSCIENCE

Figure 1. Treatment modality of OHS patients in the three groups of opium addicts, those on methadone maintenance therapy, and non-addicts

Abbreviation: CPAP: Continuous positive airway pressure; NIV: Non-invasive ventilation.

Note: A treatment modality was selected in a positive airway pressure titration study, the chi-square test was used to compare the differences between groups.

*P=0.001.

 T_{90} , and the mean value of PaCO₂ (P<0.05). In terms of different sleep stages, N1 and N2 sleep stages were significantly different between the groups (P<0.05).

Selected treatments

The selected treatment modality based on titration study is illustrated in Figure 1. As shown in Figure 1, the selected treatment modality in the three groups was significantly different (P=0.001).

4. Discussion

To our knowledge, this is the first study on PSG characteristics in patients with OHS and OA or on MMT in the Iranian population. Regarding the relatively high prevalence of all these disorders, simultaneous occurrence of OHS with OA or MMT is possible. Therefore, the evaluation of these patients with intensified problems of ventilation and sleep-disordered breathing is invaluable and useful.

We found significant differences among OHS patients in three groups, including OA, MMT and NA. The OA group had the highest score on the ESS and stop-bang questionnaire compared to other groups. The MMT group had the highest CAI, although the dominant pattern of apnea was obstructive in all groups. The longest apnea was observed in the OA group and the longest hypopnea was observed in the MMT group. Both OA and MMT groups had more sleep time with $\text{SpO}_2 <90\%$ (T₉₀), lower SpO_2 (lowest SpO_2), and higher CO_2 than the NA group. There were differences in the distribution of N1 and N2 sleep stages, but not in sleep efficiency and N3 and REM stages. Regarding selected treatment, unlike the NA group, which had a higher response rate to CPAP, the OA and MMT groups had a higher response rate to NIV.

The most important finding in our study is more striking and prolonged hypoxia in OHS patients in the OA and MMT groups. The cardiometabolic complications of sleep apnea syndrome are attributed to intermittent hypoxia and secondary activation of inflammatory pathways, the sympathetic system, and oxidative stress (Arnaud et al., 2020; Ryan et al., 2020). Therefore, this group of patients will be at risk for more severe complications and subsequent organ damage. Mediano et al. (2007) reported that OSA patients with higher ESS scores had lower mean and minimum SpO₂ in comparison to OSA patients with lower ESS scores. Similarly, in our study, the OA and MMT groups had higher ESS scores and lower minimum SpO₂ values compared to the NA group,

Variables –	Mean±SD/No. (%)			D*
	OA (n=30)	MMT (n=15)	NA (n=30)	r
Age (y)	58.22±12.21	57.28±12.92	52.5±16.8	0.28
Male	30(100)	10(66.6)	14(46.6)	0.001
BMI (kg/m²)	35.9±6.1	37.4±6.5	38.1±7.4	0.44
ESS score	17.75±12.90	12.53±4.37	10.87±5.41	0.01
Stop-bang questionnaire score	6.27±1.20	5.87±1.19	5.07±1.34	0.002
Berlin questionnaire score	7.43±2.47	6.53±2.13	6.53±2.15	0.25
Diabetes mellitus	3(10)	3(20)	8(26.6)	0.25
Hypertension	15(50)	6(40)	12(40)	0.69
Hyperlipidemia	0	3(20)	2(6.6)	0.04
Ischemic heart disease	5(16.6)	3(20)	7(23.3)	0.81
Hypothyroidism	0	0	1(3.3)	0.46

Table 1. Demographic and clinical characteristics of the three studied groups with obesity hypoventilation syndrome

NEURSSCIENCE

Abbreviaton: OA: Opium addicts; MMT: Methadone maintenance treatment; NA: Non-addicts; BMI: Body mass index; ESS: Epworth sleepiness scale.

*P<0.05.

but the mean SpO_2 was not different between the three groups (Mediano et al., 2007). This difference can be attributed to the difference between the mean ESS score in the two studies. In our study, the ESS score in the three groups was more than ten, but in the study by Mediano et al. (2007) the ESS score in one group was <10. We found the longest duration for apnea and hypopnea in the OA and MMT groups. This is similar to a previous study by Asadpour et al. who reported prolonged apnea in patients with chronic opium use (Asadpour et al., 2020).

The high prevalence of daytime hypercapnia and chronic respiratory failure in chronic opioid users has been reported previously (Rose et al., 2014). Similarly, in our study, despite the existence of hypercapnia in all patients as a diagnostic criterion of OHS, the mean $PaCO_2$ was higher in the OA and MMT groups than in the NA group. This could be due to the additional respiratory depression effect of these drugs. Hypercapnia has been attributed to adverse consequences of OSA in addition to intermittent hypoxia. Therefore, this finding emphasizes the susceptibility of these groups to more complications.

Similar to our results, in the study by Guilleminault et al. (2010) on a large group of chronic opioid users, OSA was diagnosed, and biphasic positive airway pressure (BiPAP, form of NIV) with backup rate successfully eliminated apnea, desaturation, and related symptoms (Guilleminault et al., 2010). In their study, patients with a BMI of more than 28 kg/m² were excluded, but the majority of apnea events in these patients were obstructive. Chronic opioid use is not always associated with CSA. In the mentioned study, the most effective treatment in chronic opioid users was BiPAP with a backup rate, unlike the control group that responded well to CPAP. In our study, the most effective treatment in the majority of cases in the OA and MMT groups was NIV.

Similar to the current study, in the study by Amra et al. opium addiction in OSA patients was associated with lower SpO_2 , higher CAI, lower sleep latency, and the same sleep efficiency (Amra, 2015). Unlike our study, opium-addicted patients in that study had higher AHI. The difference could be explained by studied patients, OHS in our study vs OSA in their study.

The typical effect of opioids on sleep architecture included the increase of stage N2, and the decrease of REM and N3 stages has been observed in patients after acute prescription of these drugs and this is not frequently seen in addicted patients (Kay et al., 1979). In our study, the

Variables		Mean+SD			
		OA (n=30)	MMT (n=15)	NA (n=30)	• P *
AHI		46.9±19.9	39.9±26.6	50.03±26.9	0.43
CAI		3.58±1.2	7.1±3.9	3.4±1.3	0.01
Longest apnea (s)		50.1±4.6	38.6±2.5	27.2±2.7	0.001
Longest hypopnea (s)		64.3±19.5	76.4±15.6	65.8±17.9	0.04
Mean SpO ₂ (%)		80.8±16.02	83.4±5.01	86.6±8.3	0.19
Minimum SpO ₂ (%)		72.9±11.7	67.08±12.4	76.9±12.9	0.03
T ₉₀ (%)		71.7±37.7	82.4±23.6	50.4±37.9	0.009
PaCO ₂		58.8±10.2	59.8±7.08	53.8±8.04	0.04
Sleep efficacy (%)		56.9±18.3	67.5±16.1	62.6±13.7	0.10
Sleep Latency (%)		20.16±11.57	12.47±7.1	23.3±14.6	0.15
REM latency (%)		36±6.1	18.1±3.4	52.9±7.1	0.001
Wake (%)		42.8±17.5	32.5±16.3	38.8±16.2	0.13
Sleep stage (%)	N1	26.1±8.1	20.8±4.5	38.1±8.9	0.01
	N2	22.3±4.9	57.4±8.5	13.9±1.3	0.005
	N3	6.7±1.7	12.2±1.4	8.7±0.9	0.20
	REM	2.3±0.4	1.6±0.3	2.5±0.6	0.87
HR		80.5±15.8	72.4±18.2	78.6±15.1	0.34
PLM index		4.8±0.9	4.8±1.2	3.2±0.6	

Table 2. Polysomnographic characteristics of the three studied groups with obesity hypoventilation syndrome

Abbreviation: OA: Opium addicts; MMT: Methadone maintenance treatment; NA: Non-addicts; AHI: Apnea hypopnea index; CAI: Central apnea index; SpO₂: Oxygen saturation measured by pulse oximetry; REM: Rapid eye movement; HR: Heart rate; PLM: Periodic limb movement.

*P<0.05.

duration of N2 was prolonged significantly, but the reduction of N3 and REM stages was not significant in the OA and MMT groups. These differences might be related to a relatively small sample size, especially in the MMT group.

Filiatrault et al. (2016) reported a moderate increase in CAI in chronic opioid users in a systematic review and meta-analysis of seven observational studies. They explained the possible role of confounding factors considering the observational nature of the studies (Filiatrault et al., 2016). In our study, CAI was significantly higher in the MMT group than in other groups. But in the OA group, our finding was different. This finding may be related to the obseity of the studied population. The as-

sociation between obesity and OSA has been shown in other studies (Jehan et al., 2017). Therefore, this observation in the OA group could be explained.

In another study, the authors evaluated a group of patients under treatment with opioid medications because of pain and reported a higher prevalence of CSA in them compared to the control groups, including subjects with and without pain but without opioid intake (Jungquist et al., 2012). Different results could be explained by the completely different nature of the mentioned study's population. They were prescribed controlled doses of opioids for chronic pain, but our patients were addicted to opium. Additionally, in their study, patients on MMT were excluded. In their study, there was no difference in the minimum SpO₂ between the groups. The majority of studies on patients with OA or on MMT have assessed sleep apnea syndrome. Studying patients with OHS and OA or those on MMT can be done by the evaluation of sleep characteristics in the presence of two disturbing factors, such as obesity and drugs.

Some limitations in this study should be addressed. The number of patients in all groups, especially in the MMT group was small. The gender distribution of the groups was not similar. The method of categorization was based on the self-report of patients, not laboratory tests. We could not evaluate the exact dose of opium. We did not categorize the patients regarding the dose of opium and methadone. Although all patients in the OA and MMT groups were using the stable dose of these substances, we could not exclude the effect of different doses on the results. We did not consider the duration of opium addiction or methadone treatment beyond three months, which might have affected the final results.

5. Conclusion

Based on the results of this study, AHI was similar among OHS patients in three groups of OA, MMT, and NA. The ESS and STOP-Bang questionnaire scores and the longest apnea duration were higher in the OA group than in the MMT and NA groups. On the other hand, CAI, the longest hypopnea duration, and the minimum SpO_2 , T_{90} , and $PaCO_2$ values were higher in the MMT group compared to the OA and NA groups. These results suggest the need for further studies to compare the effects of opium and methadone on sleep in OHS patients.

Ethical Considerations

Compliance with ethical guidelines

This study was approved by the Research Ethics Committee of Isfahan University of Medical Science (Code: IR.MUI.MED.REC.1399.651). Informed consent was obtained from all patients.

Funding

This research received no specific grant from any funding agency in the public, commercial, or not-for-profit sectors.

Authors' contributions

Conceptualization, methodology and supervision: Babak Amra; Visualization and validation: Forogh Soltaninejad; Formal analysis: Marjan Mansourian; Software: Mehrzad Salmasi; Data curation and writing the original draft: Hossein Narimani; Review and editing: Forogh Soltaninejad Ingo Fietze: Christoph Schöbel.

Conflict of interest

The authors declared no conflict of interest.

Acknowledgments

The authors would like to thank the patients for their participation.

References

- Aflatoonian, M. R., Sharifi, I., Hakimi Parizi, M., Fekri, A. R., Aflatoonian, B., & Sharifi, M., et al. (2014). A prospective cohort study of cutaneous leishmaniasis risk and opium addiction in south eastern Iran. *Plos One*, 9(2), e89043. [DOI:10.1371/ journal.pone.0089043] [PMID] [PMCID]
- Amra B. (2015). The effect of opium on polysomnographic findings in patients with sleep apnea [PhD thsis]. Isfahan: Isfahan University of Medical Sciences. [Link]
- Ara, A., Jacobs, W., Bhat, I. A., McCall, W. V. (2016). Sleep disturbances and substance use disorders: A bi-directional relationship. *Psychiatric Annals*, 46(7), 408-412. [DOI:10.3928/00485713-20160512-01]
- Arnaud, C., Bochaton, T., Pépin, J. L., & Belaidi, E. (2020). Obstructive sleep apnoea and cardiovascular consequences: Pathophysiological mechanisms. *Archives of Cardiovascular Diseases*, 113(5), 350–358. [DOI:10.1016/j.acvd.2020.01.003] [PMID]
- Asadpour, H., Naghibi, S. M., Rahimi, S., Sharafkhaneh, A., Afshari Saleh, L., & Rezaee Talab, F., et al. (2020). Prolonged sleep apnea in two patients with a history of opium abuse -a case report. *Iranian Journal of Otorhinolaryngology*, 32(109), 127–131. [DOI:10.1183/09031936.00009506] [PMID] [PMCID]
- Baldassarri, S. R., Beitel, M., Zinchuk, A., Redeker, N. S., Oberleitner, D. E., & Oberleitner, L. M. S., et al. (2020). Correlates of sleep quality and excessive daytime sleepiness in people with opioid use disorder receiving methadone treatment. *Sleep & Breathin*, 24(4), 1729–1737. [DOI:10.1007/s11325-020-02123-z] [PMID] [PMCID]
- Ekhtiari, H., Noroozi, A., Farhoudian, A., Radfar, S. R., Hajebi, A., & Sefatian, S., et al. (2020). The evolution of addiction treatment and harm reduction programs in Iran: A chaotic response or a synergistic diversity?. *Addiction*, 115(7), 1395– 1403. [DOI:10.1111/add.14905] [PMID]
- Filiatrault, M. L., Chauny, J. M., Daoust, R., Roy, M. P., Denis, R., & Lavigne, G. (2016). Medium increased risk for central sleep apnea but not obstructive sleep apnea in long-term opioid users: A systematic review and meta-analysis. *Journal of Clinical Sleep Medicine*, 12(4), 617–625. [DOI:10.5664/jcsm.5704] [PMID] [PMCID]

- Finlay, M., Wilson, M., Erwin, J., Hansen, D., Layton, M., & Quock, R., et al. (2020). 0730 individuals receiving methadone for medication-assisted treatment of opioid use disorder show evidence of respiratory depression. *Sleep*, 43(Supplement_1), A278. [DOI:10.1093/sleep/zsaa056.726]
- Guilleminault, C., Cao, M., Yue, H. J., & Chawla, P. (2010). Obstructive sleep apnea and chronic opioid use. *Lung*, 188(6), 459–468. [DOI:10.1007/s00408-010-9254-3] [PMID]
- Iftikhar, I. H., & Roland, J. (2018). Obesity hypoventilation syndrome. *Clinics in Chest Medicine*, 39(2), 427–436. [DOI:10.1016/j.ccm.2018.01.006] [PMID]
- Jehan, S., Zizi, F., Pandi-Perumal, S. R., Wall, S., Auguste, E., & Myers, A. K., et al. (2017). Obstructive Sleep Apnea and Obesity: Implications for Public Health. *Sleep Medicine and Disorders*, 1(4), 00019. [PMID] [PMCID]
- Jorgensen, G., Downey, C., Goldin, J., Melehan, K., Rochford, P., & Ruehland, W. (2020). An Australasian commentary on the AASM manual for the scoring of sleep and associated events. *Sleep and Biological Rhythms, 18*(3), 163-85. [DOI:10.1007/ s41105-020-00259-9]
- Jungquist, C. R., Flannery, M., Perlis, M. L., & Grace, J. T. (2012). Relationship of chronic pain and opioid use with respiratory disturbance during sleep. *Pain Management Nursing*, 13(2), 70–79. [DOI:10.1016/j.pmn.2010.04.003] [PMID]
- Kay, D. C., Pickworth, W. B., Neidert, G. L., Falcone, D., Fishman, P. M., & Othmer, E. (1979). Opioid effects on computerderived sleep and EEG parameters in nondependent human addicts. *Sleep*, 2(2), 175–191. [DOI:10.1093/sleep/2.2.175] [PMID]
- Kakazu, M. T., Soghier, I., Afshar, M., Brozek, J. L., Wilson, K. C., & Masa, J. F., et al. (2020). Weight loss interventions as treatment of obesity hypoventilation syndrome. A systematic review. *Annals of the American Thoracic Society*, *17*(4), 492–502. [DOI:10.1513/AnnalsATS.201907-554OC] [PMID]
- Le, T. A., Le, M. Q. T., Dang, A. D., Dang, A. K., Nguyen, C. T., & Pham, H. Q., et al. (2019). Multi-level predictors of psychological problems among methadone maintenance treatment patients in difference types of settings in Vietnam. *Substance Abuse Treatment, Prevention, and Policy*, 14(1), 39. [DOI:10.1186/s13011-019-0223-4] [PMID] [PMCID]
- Martel, M. O., Shir, Y., & Ware, M. A. (2018). Substance-related disorders: A review of prevalence and correlates among patients with chronic pain. *Progress in Neuro-Psychopharmacology* & *Biological Psychiatry*, 87(Pt B), 245–254. [DOI:10.1016/j.pnpbp.2017.06.032] [PMID]
- Masa, J. F., Pépin, J. L., Borel, J. C., Mokhlesi, B., Murphy, P. B., & Sánchez-Quiroga, M. Á. (2019). Obesity hypoventilation syndrome. *European Respiratory Review*, 28(151), 180097. [DOI:10.1183/16000617.0097-2018] [PMID] [PMCID]
- McPherson, M. L., Walker, K. A., Davis, M. P., Bruera, E., Reddy, A., & Paice, J., et al. (2019). Safe and appropriate use of methadone in hospice and palliative care: Expert consensus white paper. *Journal of Pain and Symptom Management*, 57(3), 635– 645.e4. [DOI:10.1016/j.jpainsymman.2018.12.001] [PMID]

- Mediano, O., Barceló, A., de la Peña, M., Gozal, D., Agustí, A., & Barbé, F. (2007). Daytime sleepiness and polysomnographic variables in sleep apnoea patients. *The European Respiratory Journal*, 30(1), 110–113. [DOI:10.1183/09031936.00009506] [PMID]
- Mubashir, T., Nagappa, M., Esfahanian, N., Botros, J., Arif, A. A., & Suen, C., et al. (2020). Prevalence of sleep-disordered breathing in opioid users with chronic pain: A systematic review and meta-analysis. *Journal of Clinical Sleep Medicine*, 16(6), 961–969. [DOI:10.5664/jcsm.8392] [PMID] [PMCID]
- Rose, A. R., Catcheside, P. G., McEvoy, R. D., Paul, D., Kapur, D., & Peak, E., et al. (2014). Sleep disordered breathing and chronic respiratory failure in patients with chronic pain on long term opioid therapy. *Journal of Clinical Sleep Medicine*, 10(8), 847–852. [DOI:10.5664/jcsm.3950] [PMID] [PMCID]
- Russolillo, A., Moniruzzaman, A., & Somers, J. M. (2018). Methadone maintenance treatment and mortality in people with criminal convictions: A population-based retrospective cohort study from Canada. *Plos Medicine*, 15(7), e1002625. [DOI:10.1371/journal.pmed.1002625] [PMID] [PMCID]
- Ryan, S., Cummins, E. P., Farre, R., Gileles-Hillel, A., Jun, J. C., & Oster, H., et al. (2020). Understanding the pathophysiological mechanisms of cardiometabolic complications in obstructive sleep apnoea: Towards personalised treatment approaches. *The European Respiratory Journal*, 56(2), 1902295. [DOI:10.1183/13993003.02295-2019] [PMID]
- Shekarchizadeh, H., Ekhtiari, H., Khami, M. R., & Virtanen, J. I. (2012). Patterns of pre-treatment drug abuse, drug treatment history and characteristics of addicts in methadone maintenance treatment in Iran. *Harm Reduction Journal*, 9, 18. [DOI:10.1186/1477-7517-9-18] [PMID] [PMCID]
- Soltaninejad, F., Sadeghi, A., & Amra, B. (2017). Compliance with continuous positive airway pressure in Persian patients with obstructive sleep apnea. *Journal of Research in Medical Sciences*, 22, 114. [DOI:10.4103/jrms.JRMS_108_17] [PMID] [PMCID]